Author's response to reviews

Title: A Comparison of a New Multinomial Stopping Rule with Stopping Rules of Fleming and Gehan in Single Arm Phase II Cancer Clinical Trials

Authors:

John R Goffin (john.goffin@jcc.hhsc.ca)
Greg R Pond (gpond@mcmaster.ca)
Dongsheng Tu (dtu@ctg.queensu.ca)

Version: 4 Date: 17 June 2011

Author's response to reviews:

Dear Editor,

We are pleased to submit our revisions. We have complied with the formatting changes requested as below.

Keywords were removed.

The footnote was removed as it was redundant. It explained abbreviations already explained in the text.

All Tables are referred to in the text.

We eliminated Table shading and unnecessary vertical and horizontal lines.

Typography was checked.

Thank you for your guidance.

Sincerely,

John Goffin